Northumbria Research Link

Citation: Lim, Michael, Whalen, Dustin, Martin, James, Mann, Paul, Hayes, Sam, Fraser, Paul, Berry, H. B. and Ouellette, D. (2020) Massive Ice Control on Permafrost Coast Erosion and Sensitivity. Geophysical Research Letters, 47 (17). e2020GL087917. ISSN 0094-8276

Published by: American Geophysical Union

URL: https://doi.org/10.1029/2020gl087917 < https://doi.org/10.1029/2020gl087917 >

This version was downloaded from Northumbria Research Link: http://nrl.northumbria.ac.uk/id/eprint/44213/

Northumbria University has developed Northumbria Research Link (NRL) to enable users to access the University's research output. Copyright © and moral rights for items on NRL are retained by the individual author(s) and/or other copyright owners. Single copies of full items can be reproduced, displayed or performed, and given to third parties in any format or medium for personal research or study, educational, or not-for-profit purposes without prior permission or charge, provided the authors, title and full bibliographic details are given, as well as a hyperlink and/or URL to the original metadata page. The content must not be changed in any way. Full items must not be sold commercially in any format or medium without formal permission of the copyright holder. The full policy is available online: http://nrl.northumbria.ac.uk/policies.html

This document may differ from the final, published version of the research and has been made available online in accordance with publisher policies. To read and/or cite from the published version of the research, please visit the publisher's website (a subscription may be required.)







RESEARCH LETTER

10.1029/2020GL087917

Key Points:

- The volumetric change at a massive ice site associated with subsidence was over 3 times greater than coastal erosion since 1994
- Passive seismic surveys have been used to map current massive ice distribution across the site in the context of these landscape dynamics
- Accounting for massive ice presence and sensitivity improves forecasts of permafrost coast erosion and inferences of carbon release rates

Supporting Information:

Supporting Information S1

Correspondence to:

M. Lim, michael.lim@northumbria.ac.uk

Citation:

Lim, M., Whalen, D., Martin, J., Mann, P. J., Hayes, S., Fraser, P., et al. (2020). Massive ice control on permafrost coast erosion and sensitivity. *Geophysical Research Letters*, 47, e2020GL087917. https://doi.org/10.1029/2020GL087917

Received 18 MAR 2020 Accepted 18 AUG 2020 Accepted article online 21 AUG 2020

©2020. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Massive Ice Control on Permafrost Coast Erosion and Sensitivity

M. Lim¹, D. Whalen², J. Martin¹, P. J. Mann¹, S. Hayes¹, P. Fraser², H. B. Berry³, and D. Ouellette⁴

¹Engineering and Environment, Ellison Building, Northumbria University, Newcastle Upon Tyne, UK, ²Natural Resources Canada, Geological Survey of Canada–Atlantic, Dartmouth, Nova Scotia, Canada, ³Department of Earth and Environmental Sciences, Dalhousie University, Halifax, Nova Scotia, Canada, ⁴Department of Civil Engineering, Schulich School of Engineering, University of Calgary, Calgary, Alberta, Canada

Abstract High overall rates of permafrost cliff retreat, coupled with spatial variability, have been accompanied by increased uncertainty over future landscape dynamics. We map long-term (>80 years) retreat of the shoreline and photogrammetrically analyze historic aerial imagery to quantify the processes at a permafrost coast site with massive ground ice. Retreat rates have been relatively constant, but topographic changes show that subsidence is a potentially critical but often ignored component of coastal sensitivity, exceeding landward recession by over three times during the last 24 years. We calibrate novel passive seismic surveys along clear and variable exposures of massive ground ice and then spatially map key subsurface layers. Combining decadal patterns of volumetric change with new ground ice variation maps enables past trends to be interpreted, future volumetric geomorphic behavior to be better constrained, and improves the assessment of permafrost coast sensitivity and the release of carbon-bearing material.

Plain Language Summary Arctic coasts are eroding faster than other coasts, in part because they contain ground ice and frozen material that is particularly sensitive to changes in temperature and the many associated processes it influences. These high erosion rates threaten coastal communities, infrastructure, and habitats, but understanding future patterns of change is complicated by the often unpredictable presence of ground ice located within coastal material and how it is responding over time. Here we examine a site on the Beaufort Sea coast where there are known large exposures of this ground ice. We used historic images to create a 3-D model of past land surface heights to calculate changes over time. These changes show large-scale lowering of the land surface over recent decades, in addition to the erosion of the coast. We then apply a new ground measurement approach to map where the ice remains. These new data are vital for understanding both future erosion processes and the subsequent release of climatically important gases and material.

1. Introduction

Arctic permafrost is found on 34% of Earth's coasts (Lantuit et al., 2012) and is estimated to release 14 teragrams of particulate organic carbon to the nearshore zone each year, equivalent to the annual Arctic riverine inputs or vertical net methane emissions from terrestrial permafrost (Fritz et al., 2017). Pan-Arctic coasts on average recede by ~0.5 m a⁻¹ (Lantuit et al., 2012), but coasts consisting of ice-rich permafrost sediments have been shown to be retreating much faster in recent decades (Radosavljevic et al., 2016), exceeding 20 m a⁻¹ in places (Solomon, 2005). Constraining erosion rates, responses, and trends is essential to mitigating the threats to Arctic communities and infrastructure (Radosavljevic et al., 2016), and establishing the resultant material fluxes from coastal erosion impacts on global carbon release and climate feedbacks (Couture et al., 2018).

Ice-rich permafrost terrain comprises a surficial "active layer," a shallow (generally <1 m) layer that thaws seasonally (Lyon et al., 2009); perennially frozen "permafrost" that can reach hundreds of meters thick (Lantuit et al., 2012) and areas of segregated ice known as "massive ice" (Mackay, 1972; Murton, 2009). Massive ice can have multiple forms and origins (Gilbert et al., 2016; Mackay, 1980) but can generally be defined as large bodies of ground ice where gravimetric moisture contents exceed 250% by weight (Pollard, 1990). This epigenetic volume expansion (formed within existing sediments as permafrost aggrades) typically results in uplift, altering local morphology and producing a landscape of domed mounds at a range of scales (Gilbert et al., 2016). Massive ice contains only trace concentrations of organic carbon but can account for over 90% of the eroding cliff face exposure where ice cored domes are exposed at the coast (Pollard, 1990).

Ground ice layers are sporadically and often unpredictably distributed (Mackay, 1972; Pollard, 1990), restricting the best estimates of volume and inferred carbon release to a standard factor to account for ice content in erosion rates (e.g., an average value of 19% used by Couture et al., 2018). The variation in massive ice presence and associated processes often remains restricted to visible exposures and boreholes (Dallimore et al., 1996), which are often limited in distribution and accessibility. Consequently, the role of massive ice in controlling erosion rates and failure processes is generally omitted from models projecting permafrost coast change (Smith et al., 2007), with notable exceptions (for example Dallimore et al., 1996). Here we analyze historic aerial imagery to establish past trends in a key massive ice type-site on the ice-rich permafrost coast of the Beaufort Sea, Arctic Canada. A sequence of volumetric changes shows that despite high rates of coastal erosion the area was dominated by the rapid collapse of a massive ice dome. Within this context we use a passive seismic survey approach to map massive ice layers for the first time, to better understand and constrain the future evolution of the site and to assess the implications for improving estimates of organic, potentially carbon-rich material release into the sea.

2. Regional Setting and Methods

The Tuktoyaktuk Coastlands, northwest Canada, are located within the southern Beaufort Sea area and are dominated by ice-rich permafrost landscapes (Rampton, 1988). This generally flat, deltaic landscape is punctuated by ice cored plateaus and domes across the tundra. Here, we focus on Peninsula Point (Figure 1a), 6 km southwest of Tuktoyaktuk, and within the Pingo Canadian Landmark (a national historic site managed by Parks Canada), it is a representative type-site for intrasedimental massive ice (Gilbert et al., 2016; Mackay & Dallimore, 1992; Murton, 2009). Compounding the dynamic erosion processes in the area (Murton, 2005), reduced sea-ice (Overeem et al., 2011) and frozen ground seasons (Laberge & Payette, 1995; Liljedahl et al., 2016), increasing storm intensity (Vermaire et al., 2013) and a relative 2.5 mm a⁻¹ sea-level rise (Hill et al., 1993) have intensified the degradation of permafrost coasts in the region.

2.1. Change Detection: Shoreline Retreat and Photogrammetric Modeling

Aerial images from the Natural Resources Canada archive (https://www.nrcan.gc.ca/maps-tools-publications/satellite-imagery-air-photos/air-photos/national-air-photo-library/9265) have been georeferenced using common features to those identifiable within a 2018 orthoimage. Check points excluded from the georeferencing process have been used to produce the error bounds in the shoreline positions. The older positions have been georeferenced from the work of Mackay and Dallimore (1992), and as neither the raw images nor the calculated registration errors were available, the highest check point error (15 m in 1985, accounting for possible image distortion, plotting accuracy, and georeferencing errors) has been assumed for the shoreline positions in 1935; 1971 and 1950 (Figure 1b). Rates of change between survey periods have been established within ESRI ArcGIS, calculated by densifying the digitized shorelines to a measurement every 10 m before computing the mean Euclidean distance between positions (Figure 1c). The shoreline positions in the aerial images may also be influenced by small tidal variations or sea-level changes over the intervening period. Therefore, although beyond the error bands used (15 m), the coastline retreat rates should be interpreted with caution.

Historic aerial imagery (10 images with at least 60% overlap) of Peninsula Point from 1994 have been photogrammetrically analyzed in Agisoft Metashape[©] using Structure from Motion processing to establish the decadal scale volumetric changes occurring at the site. The 1994 photogrammetric DEM (11 points per m²) has been compared to a 2004 lidar flight (collected at 1 point per m²) and a 2018 UAV photogrammetric survey (400 points per m²). The 2018 surface has been derived from 480 12 MP images (captured from a DJI Phantom 4 at a flying height of 70 m). Three-dimensional control points for the 1994 survey have been taken from common features with the 2004 lidar surface using areas away from any signs of activity, breaks of slope, or significant changes in topography. The large amounts of genuine change posed significant challenges to the number (6) and distribution of checkpoints achieved, selected in the same way but excluded from the registration process; in reality these points were often close to the control points used. A root mean square check point error of 0.7 m was recorded, but the 1994 photogrammetric surface was consistently





Figure 1. Site location (a) relative to the global (first inset) and regional (second inset) context. Retreat at Peninsula Point since 1935 (c) has been derived from digitized shoreline positions (b). The dashed box shows the survey area presented in Figures 4 and 5, and the black dot shows the approximate location of a borehole recording massive ice thickness, presented in Mackay and Dallimore (1992). A UTM reference scale is presented on the outer axes.

higher on the landward (assumed stable) side than in 2004 (Figure 2e). It cannot be determined whether this is a genuine surface lowering (approximately 0.5 m) from 1994 to 2004 or the result of vegetation changes for example, but the trend does not appear from 2004 to 2018. Therefore, error bounds based on the standard deviation of elevation differences in "nonchange areas" have been used in the interpretation of change. This resulted in error thresholds of ± 0.7 m in the 1994–2004 changes and ± 0.2 m in the 2004–2018 changes.

2.2. Passive Seismic Characterization of Massive Ice Spatial Variability

Passive seismic surveys have been conducted in order to map and characterize the spatial distribution of subsurface layers across the site at Peninsula Point. Tromino^{*} 3G ENGY instruments record three component accelerometer measurements of the ambient noise that is generated by microseisms from the Earth's interior and surface waves, caused principally by wave action at the shoreline (Kumar et al., 2018). The microseisms and surface wave signals propagate through the near-surface layers. Notable differences between amplitudes of horizontal and vertical motions are caused by acoustic impedance contrasts at layer boundaries and differences between shear-wave and compressional-wave velocities within each surface layer. The spectral ratio of horizontal over vertical responses exhibits resonance frequencies (fr) that can be directly linked to the layer thickness (h). The layer thicknesses are estimated by the quarter wavelength resonance approximation that was initially proposed by Joyner et al. (1981) and then later optimized for a wide variety of ground-motion metrics in diverse ground conditions (Boore, 2003) but primarily for generic rock profiles rather than frozen ground. Measurements of visible contact layer depths exposed in permafrost cliff sections adjacent to a set of control reading locations have been used to characterize relative acoustic impedance contrast signatures and subsequently to calculate shear-wave velocity (V_s) through each of the main material





Figure 2. Topographic changes at Peninsula Point from the 1994 surface (a), through 2004 (b) to 2018 (c). The degradation of the site is driven by both subsidence of a large domed area stretching across the site (profile e) and a channel area behind the cliff front (labelled in c), and by landward erosion of the cliff face (profile e).

layers (see supporting information). This has enabled the determination of two distinct high frequency signal peaks associated with specific ground layer controls. A single peak in the resonance frequency (usually around 20 Hz) denoted active layer or slurry contact directly onto massive ice and a single deep later reflector, whereas traces with a high frequency peak around 30-50 Hz (usually with additional lower frequency peaks of 1-5 Hz) represented a permafrost to ice contact. There was no validation for the ice base contact, which was not exposed in the cliff, but borehole data from Mackay and Dallimore (1992) indicate that the ice in the areas is between 10 and 20 m thick. Once the impedance contrast signatures were classified, the following formula was applied to calculate and validate layer boundary depths (*h*):

$$h = \frac{V_s}{4f_r}$$
(1)

A shear-wave velocity of 920 m s⁻¹ for the permafrost was determined from cliff outcrop data at the site and was used to derive depth to the massive ice contact (Carcione & Seriani, 1998). Validation sites have also been used to make sure that layer properties were sufficiently consistent over the survey area (see supporting





Figure 3. Surface difference models between 1994 and 2004 (a) and 2004 and 2018 (b) demonstrating the difference in volume loss through coastal erosion (material between the sequential shoreline positions) and vertical subsidence of the land surface.

information). These shear-wave velocities matched the control site surveys to known contacts but are at the lower end of what would be expected for permafrost shear-wave velocities, potentially reflecting areas of talik formation or degradation within the permafrost body, changing phase behavior, and slowing the transmittance speeds (Carcione & Seriani, 1998). Sites were selected to reflect changes in topography, usually at the top, middle, and base of a slope. Further research is required on whether transmittance speeds vary through the thaw season. The permafrost to ice contact depths, where present, have been combined with active layer probing and subtracted from surface elevations derived from the high-resolution UAV DEM. The massive ice surface has been modeled across the active areas of the site using smooth interpolation to reflect the continuous undulations of the surfaces exposed in the cliff sections.

3. Results

3.1. Patterns of Change and the Role of Massive Ice

The mapped positions of the shoreline at Peninsula Point reveal a relatively continuous retreat rate of 3.47 m a^{-1} since 1935 (Figure 1c). This rate is within the long-term range (1 m–5 m a^{-1}) for the area (Solomon, 2005) but lower than the 5 m a^{-1} rate for ice face retreat the site determined by French et al. (1983) and significantly higher than rates from 1952 (0.5–1.3 m a^{-1}) established to the west along the Yukon coastline (Lantuit & Pollard, 2008; Radosavljevic et al., 2016). Air temperatures modeled using the

Canadian Global Coupled Model ensembles were also projected to rise at a relatively constant rate over this period, amounting to an approximate 4°C increase from 1960 to 2020 (Manson & Solomon, 2007). Where excess ground ice melts the ground surface subsides and conditions change. Heat fluxes and hydrology are altered significantly (Aas et al., 2019), and ice-rich permafrost in contact with rivers and coasts undergoes thermal erosion and denudation processes (Kanevskiy et al., 2016). The shape of the coastline at Peninsula Point switched from convex in 1935 (i in Figure 1b) to concave in 1985. Early reports from the site refer to a flattish ice-cored hill (French et al., 1983), and supported by the morphological changes, it appears that the center of the massive ice was seaward of its current position (i in Figure 1b) as suggested by Mackay and Dallimore (1992). Superimposed on this wide-scale morphodynamic transition are localized responses. For example, an exposed slump backwall (iii in Figure 1b) supplied high amounts of saturated material that flowed out to form a fan (ii in Figure 1b) that resulted in a localized progradation of the coastline.

Photogrammetric analysis of historic aerial imagery has enabled characterization of long-term change at the study site (Figure 2). A notable feature has been the subsidence of a large connected dome (Figure 2d) and channel (Figure 2c) feature. Localized thresholds and feedbacks such as snow and water flow accumulation are likely to play a key role in determining subsidence rates (Aas et al., 2019; Seppälä, 2011). The sections through the different ages of surfaces (1994; 2004 and 2018) also show a high degree of consistency and progressive phased development in the trends of change. Over approximately decadal time steps, the island has subsided at differential rates to reach a similar base elevation (Figures 2d and 2e), switching its overall topography from a relatively coherent dome profile to a retrogressive thaw slump (Burn & Lewkowicz, 1990; Lantuit et al., 2012). The landward profile sequence (Figure 2e) shows that coincident to the subsidence, a seaward encroachment resulted from fans of saturated thermokarst material released into the nearshore. The fan was removed within 14 years, by which time wave erosion had truncated the toe of the slope.

Subsidence of ice centered "earth hummocks" in the area has been reported at rates from 0.2 to 1 m a^{-1} (French et al., 1983). Since 1994 the central area of the site (covered by Figure 3) has experienced 0.76 M m³ (±0.11 M m³)of subsidence, while the coastal zone directly adjacent from the subsiding area has eroded by 0.23 M m³ (±0.02 M m³). Over the survey period, the volumetric ratio of subsidence to coastal recession (defined as erosion between shoreline positions) reduced from 4.06 (482,711 m³ relative to





Figure 4. (a) Interpolation of massive ice surface variations. The ice distribution is overlain by the passive seismic measurement positions and classification. "Overburden present" includes points where the massive ice is covered by a significant (>1 m) layer of either active later and unfrozen material or active layer and permafrost. Landward (b) and shore-parallel (c) profiles highlight areas where the ice comes close to the ground surface and caused localized subsidence (at distances of 110 and 340 m in (b) and (c), respectively). A UTM grid is overlain for scale in meters.

118,844 m³ lost from the coast) between 1994 and 2004 to 2.45 (279,775 m³ relative to 114,192 m³ lost from the coast) between 2004 and 2018. Using historic images and a borehole referenced in Mackay and Dallimore (1992) for validation, locations that were massive ice dominated have subsided up to 5 times faster than the island as a whole, but the 60% reduction in the difference between subsidence and coastal recession likely reflects a reduction in the supply of massive ice over time. It is noteworthy that the borehole location (Figure 1b) in 1988 was approximately 40 m from the exposed cliff face and 10 m above sea-level, but that site is now only 3.7 m above sea-level and seaward of the exposed cliff. Accounting for both the total measured subsidence and the landward retreat of the coastline, the site has experienced an average loss of 59 m³ for each meter width strip of coastline annually. Within these landscape shifts, the mechanics of change reflect the rapid lateral expansion of slump backwalls that expose ground layers to thermoerosional processes such as insolation and precipitation, but once sufficiently insulated by a drape of failed material the vertical erosion (in particular) is greatly reduced. These switches from supply to transport limited system behavior operate over a subdecadal timeframe (Farquharson et al., 2019).





Figure 5. Surface to massive ice difference model highlighting the organic-rich layer volumes. The passive seismic signal measurement positions and classifications are overlain. A UTM grid is overlain for scale in meters.

3.2. Mapping Subsurface Structure of Massive Ice

This work demonstrates that passive seismic surveys can determine the presence and depth of massive ice within challenging permafrost coast ground conditions. Stratigraphic borehole data have shown that the base of the massive ice in the area is relatively consistent but that there is significant variability in the surface elevation of the ice (Mackay & Dallimore, 1992; Moorman, 1998). The passive seismic surveys show concentrations of massive ice rise to high points in the landscape (Figure 4a; darker purple areas). Profiles through the ice and ground surfaces demonstrate that topography is still strongly associated with massive ice presence (Figures 4b and 4c). However, areas where the ice rises close to the surface, which have experienced high recent rates of subsidence, are beginning to invert this pattern with localized subsided lows where near-surface ice has melted (Figure 4b). The channel feature that has subsided significantly over the last few decades (Figure 2c) is still underlain by near-surface massive ice and thus remains susceptible to further subsidence through local processes (Aas et al., 2019). The undulating surface of the ice appears to overlie a larger north-south trend as the ice contact dips to the south of the island, reflecting an increasing dominance of permafrost in the ground layers and potentially a reduced sensitivity to rapid subsidence (Obu et al., 2017). The spatial variability of ground ice and its direct impact on thaw-induced surface deformation has recently been demonstrated in both small scale (an $11 \text{ m} \times 10 \text{ m}$ plot) field experiments (Wagner et al., 2018) and investigation of localized thermokarst signals (Paquette et al., 2020).

The wider impacts of massive ice presence and sensitivity to change are related to both the direct release of carbon bearing material (Couture et al., 2018) and to the indirect changes to the exposure of highly labile, nutrient-rich, and bio-available material across retrogressive thaw slump (Abbott & Jones, 2015) and near-shore regions (Tanski et al., 2019). As an ice-rich area sequences through rapid subsidence and backwall recession it produces an increasing transition zone that remains largely unvegetated while the supply of meltwater continues (Kokelj & Burn, 2003). Gas emissions from these retrogressive slump areas and rapid thaw areas have the potential to cause a disproportionate release of carbon, particularly in methane resulting from ground saturation (Turetsky et al., 2020). It is therefore essential to better identify the spatial variation



of massive ice and to constrain the temporal responses of these climate sensitive areas as they transition through phases of dynamic mass wasting to ice exhaustion, stabilization, and revegetation (Farquharson et al., 2019). By subtracting our modeled massive ice layer from the ground surface, a spatial map of the remaining organic-rich material present at the site (1.1 M m³ of the total 1.4 M m³ within the surveyed area) has been produced (Figure 5). The overall massive ice content within the ground is comparable to values used elsewhere (22% compared with an average of 19% used in Couture et al., 2018, on the Yukon Coastal Plain), but the spatial variability ranges from comprising effectively all of the above sea-level ground profile to being absent from it. These approaches allow spatial estimates of contributions from ice melt relative to permafrost thaw to be partitioned.

4. Conclusions

Massive ice collapse remains a critical but poorly quantified driver of ice rich permafrost coast erosion processes and as such is a threat to Arctic ecosystems, infrastructure, and biogeochemical cycles (O'Neill et al., 2019). Taking an ice-rich permafrost type-site as an example, with clear and variable exposures of massive ground ice, we set the long-term erosion trends in context and use photogrammetric analyses of historic imagery to provide new quantitative data on decadal scale geomorphic processes. A relatively constant rate of change has been derived from shoreline mapping, which may reflect the influence of a sustained presence of massive ice at the site; however, significant doubt is cast on the accuracy and validity of this metric for understanding and predicting geomorphic responses under changing conditions. The volumetric change within the surveyed area associated with subsidence has been over 3 times greater than the volumes lost to coastal erosion since 1994. Passive seismic mapping of the site has identified the remaining ice surface contact zone and revealed several remaining near-surface ice masses that are likely to be highly vulnerable to future subsidence (Farquharson et al., 2019). By accounting for massive ice variability, both future sensitivity to change and potential constituent fluxes from terrestrial and nearshore environments can be better accounted for.

Data Availability Statement

Lim, M., & Martin, J. (2020). Geophysical survey of Peninsula Point, NWT, Canada, using Tromino 3G ENGY instruments, 2019-07-31-2019-08-03 (Version 1.0) [Data set]. UK Polar Data Centre, Natural Environment Research Council, UK Research & Innovation (https://doi.org/10.5285/8CDF7941-F13B-4237-B921-05A6C6CDE40A).

References

- Aas, K. S., Martin, L., Nitzbon, J., Langer, M., Boike, J., Lee, H., et al. (2019). Thaw processes in ice-rich permafrost landscapes represented with laterally coupled tiles in a land surface model. *The Cryosphere*, 13(2), 591–609. https://www.the-cryosphere.net/13/591/2019/, https://doi.org/10.5194/tc-13-591-2019
- Abbott, B. W., & Jones, J. B. (2015). Permafrost collapse alters soil carbon stocks, respiration, CH₄, and N₂O in upland tundra. *Global Change Biology*, *21*(12), 4570–4587. https://doi.org/10.1111/gcb.13069
- Boore, D. M. (2003). Simulation of ground motion using the stochastic method. *Pure and Applied Geophysics*, 160(3), 635–676. https://doi. org/10.1007/PL00012553
- Burn, C. R., & Lewkowicz, A. G. (1990). Canadian landform examples 17 retrogressive thaw slumps. The Canadian Geographer/Le Géographe canadien, 34(3), 273–276. https://doi.org/10.1111/j.1541-0064.1990.tb01092.x
- Carcione, J. M., & Seriani, G. (1998). Seismic and ultrasonic velocities in permafrost. *Geophysical Prospecting*, 46(4), 441–454. https://doi. org/10.1046/j.1365-2478.1998.1000333.x
- Couture, N. J., Irrgang, A., Pollard, W., Lantuit, H., & Fritz, M. (2018). Coastal erosion of permafrost soils along the Yukon coastal plain and fluxes of organic carbon to the Canadian Beaufort Sea. *Journal of Geophysical Research: Biogeosciences*, 123, 406–422. https://doi.org/ 10.1002/2017JG004166
- Dallimore, S. R., Wolfe, S. A., & Solomon, S. M. (1996). Influence of ground ice and permafrost on coastal evolution, Richards Island, Beaufort Sea coast, N.W.T. *Canadian Journal of Earth Sciences*, 33(5), 664–675. https://doi.org/10.1139/e96-050
- Farquharson, L. M., Romanovsky, V. E., Cable, W. L., Walker, D. A., Kokelj, S. V., & Nicolsky, D. (2019). Climate change drives widespread and rapid thermokarst development in very cold permafrost in the Canadian High Arctic. *Geophysical Research Letters*, 46, 6681–6689. https://doi.org/10.1029/2019GL082187
- French, H. M., Heginbottom, J. A., & Phenomena, I. G. U. C. o. t. S. o. P (1983). Guidebook to permafrost and related features of the northern Yukon territory and Mackenzie delta. Canada: Division of Geological & Geophysical Surveys, Department of Natural Resources, State of Alaska.
- Fritz, M., Vonk, J. E., & Lantuit, H. (2017). Commentary: Collapsing Arctic coastlines. Nature Climate Change, 7(1), 6–7. https://doi.org/ 10.1038/nclimate3188
- Gilbert, G. L., Kanevskiy, M., & Murton, J. B. (2016). Recent advances (2008–2015) in the study of ground ice and cryostratigraphy. *Permafrost and Periglacial Processes*, 27(4), 377–389. https://doi.org/10.1002/ppp.1912

Acknowledgments

The authors thank and acknowledge the support of the NERC Arctic office UK-Canada Bursary scheme, the Inuvik and Tuktoyaktuk Hunters, and Trappers Committee and Parks Canada (Permit #: PCL-2018-2701-02), and we thank Dr. Erin Trochim and Professor Scott Lamoureux for their insightful and helpful comments.



- Hill, P. R., Héquette, A., & Ruz, M.-H. (1993). Holocene sea-level history of the Canadian Beaufort shelf. Canadian Journal of Earth Sciences, 30(1), 103–108. https://doi.org/10.1139/e93-009
- Joyner, W. B., Warrick, R. E., & Fumal, T. E. (1981). The effect of quaternary alluvium on strong ground motion in the Coyote Lake, California, earthquake of 1979. Bulletin of the Seismological Society of America, 71, 1333–1349.
- Kanevskiy, M., Shur, Y., Strauss, J., Jorgenson, T., Fortier, D., Stephani, E., & Vasiliev, A. (2016). Patterns and rates of riverbank erosion involving ice-rich permafrost (yedoma) in northern Alaska. *Geomorphology*, 253, 370–384. http://www.sciencedirect.com/science/article/pii/S0169555X15301872, https://doi.org/10.1016/j.geomorph.2015.10.023
- Kokelj, S. V., & Burn, C. R. (2003). Ground ice and soluble cations in near-surface permafrost, Inuvik, Northwest Territories, Canada. Permafrost and Periglacial Processes, 14(3), 275–289. https://doi.org/10.1002/ppp.458
- Kumar, M., Hart, J., & Prakash, N. (2018). Application of passive seismic in determining overburden thickness: North West Zambia. ASEG Extended Abstracts, 2018(1), 1–8. https://doi.org/10.1071/ASEG2018abT4_3D
- Laberge, M.-J., & Payette, S. (1995). Long-term monitoring of permafrost change in a Palsa Peatland in northern Quebec, Canada: 1983– 1993. Arctic and Alpine Research, 27(2), 167–171. https://doi.org/10.2307/1551898
- Lantuit, H., Overduin, P. P., Couture, N., Wetterich, S., Are, F., Atkinson, D., et al. (2012). The Arctic coastal dynamics database: A new classification scheme and statistics on Arctic permafrost coastlines. *Estuaries and Coasts*, 35(2), 383–400. https://doi.org/10.1007/s12237-010-9362-6
- Lantuit, H., & Pollard, W. H. (2008). Fifty years of coastal erosion and retrogressive thaw slump activity on Herschel Island, southern Beaufort Sea, Yukon Territory, Canada. *Geomorphology*, 95(1-2), 84–102. http://www.sciencedirect.com/science/article/pii/ S0169555X07001705, https://doi.org/10.1016/j.geomorph.2006.07.040
- Liljedahl, A. K., Boike, J., Daanen, R. P., Fedorov, A. N., Frost, G. V., Grosse, G., et al. (2016). Pan-Arctic ice-wedge degradation in warming permafrost and its influence on tundra hydrology. *Nature Geoscience*, 9(4), 312–318. https://doi.org/10.1038/ngeo2674
- Lyon, S. W., Destouni, G., Giesler, R., Humborg, C., Mörth, M., Seibert, J., et al. (2009). Estimation of permafrost thawing rates in a sub-arctic catchment using recession flow analysis. *Hydrology and Earth System Sciences*, 13(5), 595–604. https://www.hydrol-earth-systsci.net/13/595/2009/, https://doi.org/10.5194/hess-13-595-2009
- Mackay, J. R. (1972). The world of underground ice. Annals of the Association of American Geographers, 62(1), 1–22. https://doi.org/ 10.1111/j.1467-8306.1972.tb00839.x
- Mackay, J. R. (1980). The origin of hummocks, western Arctic coast, Canada. Canadian Journal of Earth Sciences, 17(8), 996–1006. https:// doi.org/10.1139/e80-100
- Mackay, J. R., & Dallimore, S. R. (1992). Massive ice of the Tuktoyaktuk area, western Arctic coast, Canada. *Canadian Journal of Earth Sciences*, 29(6), 1235–1249. https://doi.org/10.1139/e92-099
- Manson, G. K., & Solomon, S. M. (2007). Past and future forcing of Beaufort Sea coastal change. Atmosphere-Ocean, 45(2), 107–122. https:// doi.org/10.3137/ao.450204
- Moorman, B. (1998). The development and preservation of tabular massive ground ice in permafrost regions. (doctor of philosophy), Carleton University, Ontario.
- Murton, J. B. (2005). Ground-ice stratigraphy and formation at North Head, Tuktoyaktuk Coastlands, western Arctic Canada: A product of glacier-permafrost interactions. Permafrost and Periglacial Processes, 16(1), 31–50. https://doi.org/10.1002/ppp.513
- Murton, J. B. (2009). Stratigraphy and palaeoenvironments of Richards Island and the eastern Beaufort continental shelf during the last glacial-interglacial cycle. *Permafrost and Periglacial Processes*, 20(2), 107–125. https://doi.org/10.1002/ppp.647
- Obu, J., Lantuit, H., Myers-Smith, I., Heim, B., Wolter, J., & Fritz, M. (2017). Effect of terrain characteristics on soil organic carbon and total nitrogen stocks in soils of Herschel Island, Western Canadian Arctic. *Permafrost and Periglacial Processes*, 28(1), 92–107. https://doi.org/ 10.1002/ppp.1881
- O'Neill, H. B., Wolfe, S. A., & Duchesne, C. (2019). New ground ice maps for Canada using a paleogeographic modelling approach. *The Cryosphere*, *13*(3), 753–773. https://www.the-cryosphere.net/13/753/2019/, https://doi.org/10.5194/tc-13-753-2019
- Overeem, I., Anderson, R. S., Wobus, C. W., Clow, G. D., Urban, F. E., & Matell, N. (2011). Sea ice loss enhances wave action at the Arctic coast. *Geophysical Research Letters*, 38, L17503. https://doi.org/10.1029/2011GL048681
- Paquette, M., Rudy, A. C. A., Fortier, D., & Lamoureux, S. F. (2020). Multi-scale site evaluation of a relict active layer detachment in a high Arctic landscape. *Geomorphology*, 359, 107159. http://www.sciencedirect.com/science/article/pii/S0169555X20301318, https://doi.org/ 10.1016/j.geomorph.2020.107159
- Pollard, W. H. (1990). The nature and origin of ground ice in the Herschel Island area, Yukon Territory. Proceedings, Fifth Canadian Permafrost Conference, Québec, 23–30.
- Radosavljevic, B., Lantuit, H., Pollard, W., Overduin, P., Couture, N., Sachs, T., et al. (2016). Erosion and flooding—Threats to coastal infrastructure in the Arctic: A case study from Herschel Island, Yukon Territory, Canada. *Estuaries and Coasts*, *39*(4), 900–915. https://doi.org/10.1007/s12237-015-0046-0
- Rampton, V. N. (1988). Quaternary geology of the Tuktoyaktuk Coastlands, Northwest Territories. Memoir, (423).
- Seppälä, M. (2011). Synthesis of studies of Palsa formation underlining the importance of local environmental and physical characteristics. *Quaternary Research*, 75(2), 366–370. https://www.cambridge.org/core/article/synthesis-of-studies-of-palsa-formation-underlining-theimportance-of-local-environmental-and-physical-characteristics/7976A600DFC3655E938446430657DB99, https://doi.org/10.1016/j. yqres.2010.09.007
- Smith, L. C., Sheng, Y., & MacDonald, G. M. (2007). A first pan-Arctic assessment of the influence of glaciation, permafrost, topography and peatlands on northern hemisphere lake distribution. *Permafrost and Periglacial Processes*, 18(2), 201–208. https://doi.org/10.1002/ ppp.581
- Solomon, S. M. (2005). Spatial and temporal variability of shoreline change in the Beaufort-Mackenzie region, northwest territories, Canada. *Geo-Marine Letters*, 25(2-3), 127–137. https://doi.org/10.1007/s00367-004-0194-x
- Tanski, G., Wagner, D., Knoblauch, C., Fritz, M., Sachs, T., & Lantuit, H. (2019). Rapid CO₂ release from eroding permafrost in seawater. Geophysical Research Letters, 46, 11,244–11,252. https://doi.org/10.1029/2019GL084303
- Turetsky, M. R., Abbott, B. W., Jones, M. C., Anthony, K. W., Olefeldt, D., Schuur, E. A. G., et al. (2020). Carbon release through abrupt permafrost thaw. *Nature Geoscience*, 13(2), 138–143. https://doi.org/10.1038/s41561-019-0526-0
- Vermaire, J. C., Pisaric, M. F. J., Thienpont, J. R., Courtney Mustaphi, C. J., Kokelj, S. V., & Smol, J. P. (2013). Arctic climate warming and sea ice declines lead to increased storm surge activity. *Geophysical Research Letters*, 40, 1386–1390. https://doi.org/10.1002/grl.50191
 Warmar A. M. Lindeev, N. L. Deu, S. Colvin, A. Saori, S. Williams, C. et al. (2018). Permafraet dogradation and subsidence observations.
- Wagner, A. M., Lindsey, N. J., Dou, S., Gelvin, A., Saari, S., Williams, C., et al. (2018). Permafrost degradation and subsidence observations during a controlled warming experiment. *Scientific Reports*, 8(1), 10908. https://doi.org/10.1038/s41598-018-29292-y